Author’s response to reviews

Title: Development of the Informed Choice in Mammography Screening Questionnaire (IMQ): Factor structure, reliability, and validity

Authors:

Maren Reder (maren.reder@uni-bielefeld.de)
Eva-Maria Berens (eva-maria.berens@uni-bielefeld.de)
Jacob Spallek (jacob.spallek@b-tu.de)
Petra Kolip (petra.kolip@uni-bielefeld.de)

Version: 2 Date: 20 May 2018

Author’s response to reviews:

Dear Dr. Ragen,

Thank you for giving us the possibility to submit a revised version of our manuscript. The manuscript now includes the Results, Discussion, and Conclusion with the related figures and tables.